ANSWERING REVIEWERS

Name of journal: *World Journal of Virology*

Manuscript NO: 71202

Title: CARDIOPULMONARY CHANGES IN PATIENTS WITH SICKLE CELL ANEMIA: A SYSTEMATIC REVIEW

Reviewer’s code: 05263148

Position: Peer review

This is a well written article backed by excellent literature search and study selection based on PRISMA guidelines. However, a table summarizing the included studies with the findings would have been of great help for the readers. We are glad for your comments and suggestion on our manuscript. Your evaluation was extremely important to improve the quality of this manuscript. Aiming to attend to your suggestions, the following changes were made:

1. We have added a table summarizing the results obtained by the present study, as highlighted at the end of the manuscript in **green**.

Also, we changed the manuscript title from “CARDIOPULMONARY CHANGES IN PATIENTS WITH **FALCIFORM ANEMIA**: A SYSTEMATIC REVIEW” for “CARDIOPULMONARY CHANGES IN PATIENTS WITH **SICKLE CELL ANEMIA**: A SYSTEMATIC REVIEW” for a better understanding of the reader, since it is the most correctly used translation.

Best regards,

Fabrício Freire de Melo
Reviewer’s name: Fei-Yan Lin

Position: Science Editor

Dear reviewer,

Thank you for your evaluation and comments on our manuscript. Aiming to attend to your suggestions, the following changes were made:

1. We provided the original figure documents;

2. Also, we changed the manuscript title from “CARDIOPULMONARY CHANGES IN PATIENTS WITH FALCIFORM ANEMIA: A SYSTEMATIC REVIEW” for “CARDIOPULMONARY CHANGES IN PATIENTS WITH SICKLE CELL ANEMIA: A SYSTEMATIC REVIEW” for a better understanding of the reader, since it is the most correctly used translation.

Best regards,

Fabrício Freire de Melo.